

24 April 2022

Respected Sir/Madam

Reference#: 75806

Title: Potassium-competitive acid blockers and gastroesophageal reflux disease

I am hereby giving points response to your comments;

Reviewer #1

Comment #1: It will be better if the authors could make the content more concise as there were many literature details in all manuscript. Summarized table of these literature or studies might be helpful.

Author answer: I already added a table as your suggestion.

Comment #2: Figure and table are also needed references.

Author answer: I already added references as your suggestion.

Comment #3: Moreover, is there any study about P-CABs in children? To maximize the audience for this manuscript, application or any study in children population will be merit.

Author answer: Unfortunately, there is no study regarding P-CAB in children.

Reviewer #2

Comment #1: The author's own opinion was not enough in this review.

Author answer: I already added a paragraph regarding your suggestion.

Comment #2: Some evidence-based proof such as meta-analysis, systematic review.

Author answer: I already have several meta-analysis and systematic reviews in the manuscript.

Language polishing requirement:

Comment #1: Authors are requested to send their revised manuscript to a professional English language editing company or a native English-speaking expert to polish the manuscript further. When the authors submit the subsequent polished manuscript to us, they must provide a new language certificate along with the manuscript.

Author answer: I already submitted the manuscript to be edited by [Filipodia: Science Management & Communication Experts](#) and already received the English Language Certificate.

Science editor:

Comment #1: Please provide documents following the requirements in the journal's Guidelines for manuscript type and related ethics:

(1) Conflict-of-Interest Disclosure Form ; (2) Copyright License Agreement.

Author answer: I already added the documents regarding your suggestion.

Best regards,

W. Leowattana

Assoc.Prof.Dr.Wattana Leowattana

Tawitthep Leowattana

Dr.Tawitthep Leowattana